Weekly Status Week of November 6 – November 12, 2008

Day of Year 311 - 317
Mission Day 2915 - 2921
Earth Observing-One (EO-1) General

EO-1 has continued to perform a 2 week long experiment that started 27 October 2008 and ended 9 November 2008 to characterize the dark current drift on the ALI instrument for use by the Landsat Data Continuity Mission development team.  Consequently, there were only 21 Data Collection Events (DCEs) scheduled this week most of which occurred between Nov. 10 - Nov. 12.  

INSTRUMENTS

All instruments operated nominally this week.

Instrument decontamination cycles:
· ALI performed out-gassing on Nov. 11 from 00:30z to 15:20z.

· HSI performed de-icing on Nov. 11 from 00:20z to 15:10z.

EO-1 Spacecraft Subsystems

Command and Data Handling (C&DH)

Continued to experience problems playing back engineering data from the Solid-State-Recorders. Engineering data is being received during real-time contacts and for all science imaging events. 

Technology Activities

In addition to providing ongoing science data collection, the EO-1 extended mission supports on-orbit testbed activities for advanced technology and hyperspectral research.  The status of various validation efforts is contained in the following paragraphs.

Sensor Web & Virtual Observatory Demonstrations

A teleconference between the Advanced Information System Technology (AIST) Sensor Web collaborators was held on Thursday, November 6.  The following items of interest were discussed:

· The upcoming ESTO AIST final review for the Supplemental Award was discussed and the schedule for the team’s input of slides and text was set as Nov. 14.

· Smoke Web Coverage Service (WCS), to be based on data from the UAS, Hyperion, and BlueSky smoke prediction model, is to trigger EO-1 automatically on an on-going basis when wildfires are present.
· Dave Smithbauer, Pat Capellaere, and Dan Mandl discussed various wildfire burn products (Normalized Burn Ratio, Burn Scar, NDVI, Burn Index, Thermal Summary, and Active Fires) to finalize the algorithm weight and color schemes for each product.
· There is a need to generate Hyperion NDVI products.  More analysis is needed to distinguish between Hyperion and ALI Level 2 products to identify which products actually are available vs which ones still need to be implemented.
· By August 2009, target is to complete fire product algorithms and validations according to above listing.
· Dan Mandl is to forward Fire Potential index developed by Dar Roberts for HyspIRI to the entire team.
· Status of Stefan Falke’s smoke classifier needs to be determined.

· University of Maryland MODIS  water mask work to be discussed at GSFC on Nov. 13 at 1:00.

· Discussion with Bruce Trout on on-board processing to be held Nov. 13 in am.

A teleconference between the flood early warning sensor web collaborators was held on Tuesday, November 11.  The following items of interest were discussed:

· Automatic creation of Envisat flood maps and MODIS flood maps using the Bob Adler’s TRMM based flood potential map as a trigger

Stu Frye has been in contact with the World Bank concerning integrating their risk assessment into the GEOSS Pilot.  A meeting has been scheduled for Nov. 13.  

We received a request from Kenya, Africa on Nov. 12 to image flooding along the North shore of Lake Victoria.  The team used the Campaign Manager to initiate a request for the image to be taken early on Nov. 13.
Autonomous Sciencecraft Experiment (ASE)

The ASE controlled EO-1 all week.  

EO-1 MISSION OPERATIONS CENTER

Real-Time

Spacecraft state of health is currently nominal.

Solar array characterization performed on November 11, 2008 at 07:41z.

EO-1 has completed conducting calibration studies to measure the impact of imaging frequency on ALI sensor noise:
· EO-1 performed a DCE every other day between Nov. 3 and Nov. 9.  Normal operations resumed on Nov. 10.
Primary T&C machine:
No change from last report.

Mission Planning

ASPEN upgrade:

No change from last report.
Flight Dynamics

New Lunar calibration design: 
No change from last report.

Momentum management tests:
· FOT and JPL have conducted tests to better understand the behavior of the reaction wheels during DCE related slews.

· As each scene is targeted the ACS will command a bias to each wheel (x, y and z).  This bias nominally should be 1000rpm.  In the case of multiple collects, the scenes speed is determined by the effects of the previous bias command issued.

· Momentum management tests were conducted on the following dates:
· Nov. 6 at 15:50z - 16:30z
· Nov. 6 at 19:00z - 19:40z
· Nov. 7 at 18:05z - 18:55z
· Nov. 8 at 02:15z - 02:55z
· Nov. 8 at 05:15z - 06:10z
· Nov. 8 at 08:40z - 09:25z
· The predicted values analyzed for these tests are from simulated duals, triple, etc. collects (but performed with actual spacecraft slews in flight) on Nov. 6-8.  For each sequence of scenes in a given orbit, the final scene was the targeted at the optimal wheel speed.

· The difference from predicted to actual range from 0-300 rpm for the current bias scheme.  The differences from the modified bias scheme fall in this range. 

· The analysis concludes that it is possible to use the modified bias scheme for operations.
System Administration

New Front End Data Systems (FEDS) upgrade:

· Systems administrator and the FEDS engineer is currently working on upgrading the EO-1 front end systems.

· Systems will be hosted on newer hardware and operating systems.

GROUND AND SPACE NETWORK

Station Downtimes

McMurdo Ground Station (MGS), McMurdo, Antarctica:

No change from last report.
Operational Discrepancies

· No major problems to report.
Upcoming Events

Lunar Calibration:

· An all instrument nominal lunar calibration will be performed on Thursday, Nov. 13th during the 16:28z umbra.

· A modified lunar calibration will be performed three orbits following the nominal calibration.

MLT control maneuvers:

· FOT will conduct inclination maneuvers to increase MLT to 10am.

· First maneuver will begin at 14:02:06z on Dec 2, 2008.

· Second maneuver will begin at 13:47:13z on Dec4, 2008.

· Both maneuvers will be 400 sec. in duration.

Imagery Status

Scenes and Engineering Cals planned for week of November 6 – November 12, 2008              21
Total scenes and engineering calibrations planned for entire mission – approximately 42,778
Total Scenes:  ALI scenes in the Level 0 archive            38,659 (as of November 12, 2008)

                        Hyperion scenes in the Level 0 archive    38,409
Publications and Presentations Status 

404 publications 

284 external presentations

53 articles and press releases


Page 3 of 4

